Author’s response to reviews

Title: Epidemiological and clinical profiles of respiratory syncytial virus infection in hospitalized neonates in Suzhou, China

Authors:

Lianghua Lu (adolfhua@126.com)
Yongdong Yan (yyd3060@126.com)
Bin Yang (Bin_Yang@126.com)
Zhihui Xiao (Zhihui_xiao@163.com)
Xing Feng (Xing_Feng@163.com)
Yuqing Wang (wang_yu_qing@163.com)
Wei Ji (szdxjiwei@126.com)
Maximillion Mize (Maximillion.Mize@gmail.com)
Chuangli Hao (Chuangli_Hao@163.com)
Zhengrong Chen (chen_zheng_rong@163.com)

Version: 2 Date: 27 Aug 2015

Author’s response to reviews:

Thanks for your advice again. We would like to express our sincere gratitude to the section editor and reviewers for their constructive and positive comments.

Reviewer reports:

Reviewer #1: The revised manuscript has adequately address my concerns. I have no further comments for the authors.

Response: Thanks for your positive comments.

Reviewer #2: The authors have responded appropriately. There is still need for few spelling and grammatical corrections.
Response: This manuscript has been modified by a native-English speaker who works as PhD in Department of Cell-biology and Immunology, Health Science Center at Fort worth, University of North Texas. Thanks.

-------------------
Editorial Requests
-------------------
1. We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Response: This manuscript has been modified by a native-English speaker who works as PhD in Department of Cell-biology and Immunology, Health Science Center at Fort worth, University of North Texas. I hope this language editing meet the journal criterion. Thanks.

2. Re-name the Abstract Objectives as Background and include additional context

Response: Yes, we have changed “Objectives” to “Background” and added relative context.

3. Include all author e-mails on the title page

Response: Yes, we have added all author e-mails on the title page.

4. Include Abstract heading in the main ms

Response: Yes, It has been revised. See in text.

5. Include Keywords after the Abstract

Response: Yes, we have moved the keywords to the Abstract section.

6. Clarify in the Methods if consent was informed and also if it was written or oral

Response: Written consent forms were obtained from the guardians for all children enrolled. We have modified it in text.

7. Clarify in the AC the roles played by authors L. L. and Y. Y. in the conception or design of the work; or the acquisition, analysis, or interpretation of data for the work in order to ensure that all the contributors currently listed do meet the ICMJE guidelines for
Response: Yes, L.L. And Y.Y. wrote the main manuscript text, collected and interpreted data.